Author's response to reviews

Title: Fusion of green fluorescent protein to the C-terminus of granulysin alters its intracellular localization in comparison to the native molecule

Authors:

   Dennis A. Hanson (hansonda@u.washington.edu)
   Steven F. Ziegler (sziegler@benaroyaresearch.org)

Version: 2 Date: 9 September 2004

Author's response to reviews:

The manuscript was changed according to the e-mail sent by Christian Pfeffer on 8Sep2004. In specific:
1. The country, U.S.A., was added to the title page for the author's address.
2. Page numbers in the references were changed to full descriptions (e.g. 223-227 not 223-7).
3. A space was inserted directly after the colon and before the page numbers in the references.
4. The figure was altered to minimize the surrounding white space.